

## G OPEN ACCESS

**Citation:** Kim J-H, Choi J, Kim C, Pak GD, Parajulee P, Haselbeck A, et al. (2024) Mapping the incidence rate of typhoid fever in sub-Saharan Africa. PLoS Negl Trop Dis 18(2): e0011902. https://doi.org/10.1371/journal.pntd.0011902

**Editor:** Qu Cheng, Huazhong University of Science and Technology Tongji Medical College, CHINA

Received: April 26, 2023

Accepted: January 4, 2024

Published: February 26, 2024

**Copyright:** © 2024 Kim et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data and computer codes to generate the results are on Github (https://github.com/kimfinale/ TyphoidMapping).

Funding: This work was supported, in whole or in part, by Gavi, the Vaccine Alliance, and the Bill & Melinda Gates Foundation, via the Vaccine Impact Modelling Consortium (Grant Number OPP1157270 / INV-009125) to J.-H.K and the Severe Typhoid Fever in Africa Program (Grant Number OPP1127988) and Typhoid Fever Surveillance in Africa Program (Grant Number **RESEARCH ARTICLE** 

# Mapping the incidence rate of typhoid fever in sub-Saharan Africa

Jong-Hoon Kim<sup>1</sup>\*, Jungsoon Choi<sup>2</sup>, Chaelin Kim<sup>1</sup>, Gi Deok Pak<sup>1</sup>, Prerana Parajulee<sup>1</sup>, Andrea Haselbeck<sup>1</sup>, Se Eun Park<sup>1,3</sup>, Vittal Mogasale<sup>4#</sup>, Hyon Jin Jeon<sup>1,5,6</sup>, Annie J. Browne<sup>7</sup>, Ellis Owusu-Dabo<sup>7</sup>, Raphaël Rakotozandrindrainy<sup>8</sup>, Abdramane Soura Bassiahi<sup>9</sup>, Mekonnen Teferi<sup>10</sup>, Octavie Lunguya-Metila<sup>11,12</sup>, Christiane Dolecek<sup>13</sup>, Virginia E. Pitzer<sup>14,15</sup>, John A. Crump<sup>16</sup>, Simon I. Hay<sup>17,18</sup>, Florian Marks<sup>1,5,8,19</sup>

1 International Vaccine Institute, Seoul, Republic of Korea, 2 Department of Mathematics, Hanyang University, Seoul, Republic of Korea, 3 Graduate School of Public Health, Yonsei University, Seoul, Republic of Korea, 4 International Vaccine Institute, Seoul, Republic of Korea, 5 Cambridge Institute of Therapeutic Immunology and Infectious Disease, University of Cambridge School of Clinical Medicine, Cambridge Biomedical Campus, Cambridge, United Kingdom, 6 Madagascar Institute for Vaccine Research, University of Antananarivo, Antananarivo, Madagascar, 7 Malaria Atlas Project, Telethon Kids Institute, Perth, Australia, 8 School of Public Health, Kwame Nkrumah University of Science and Technology, Laing Building Complex J.W. Acheampong CI, Kumasi, Ghana, 9 Institut Supérieur des Sciences de la Population, Ouagadougou, Burkina Faso, 10 Armauer Hansen Research Institute, ALERT Compound Zenebework, Addis Ababa, Ethiopia, 11 Department of Microbiology, Institut National de Recherche Biomédicale, Kinshasa, Democratic Republic of Congo, 12 Department of Medical Biology, Microbiology Service, University Teaching Hospital, Ave De L'hopital, Kinshasa, Democratic Republic of the Congo, 13 Centre for Tropical Medicine and Global Health, Nuffield Department of Medicine, University of Oxford, Oxford, United Kingdom, 14 Department of Epidemiology of Microbial Diseases, New Haven, Connecticut, United States of America, 15 Yale Institute for Global Health. New Haven. Connecticut. United States of America. 16 Centre for International Health. Division of Health Sciences, University of Otago, Dunedin, New Zealand, 17 Institute for Health Metrics and Evaluation (IHME), University of Washington, Seattle, Washington, United States of America, 18 Department of Health Metrics Sciences, University of Washington, Seattle, Washington, United States of America, 19 Heidelberg Institute of Global Health, University of Heidelberg, Heidelberg, Germany

¤ Current address: Department of Health Systems Governance and Financing, World Health Organization, Geneva, Switzerland

\* jonghoon.kim@ivi.int

## Abstract

## Background

With more than 1.2 million illnesses and 29,000 deaths in sub-Saharan Africa in 2017, typhoid fever continues to be a major public health problem. Effective control of the disease would benefit from an understanding of the subnational geospatial distribution of the disease incidence.

## Method

We collated records of the incidence rate of typhoid fever confirmed by culture of blood in Africa from 2000 to 2022. We estimated the typhoid incidence rate for sub-Saharan Africa on 20 km × 20 km grids by exploring the association with geospatial covariates representing access to improved water and sanitation, health conditions of the population, and environmental conditions.

OPPGH5231) to F.M. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Results

We identified six published articles and one pre-print representing incidence rate estimates in 22 sites in 2000–2022. Estimated incidence rates showed geospatial variation at subnational, national, and regional levels. The incidence rate was high in Western and Eastern African subregions followed by Southern and Middle African subregions. By age, the incidence rate was highest among 5–14 yo followed by 2–4 yo, > 14 yo, and 0–1 yo. When aggregated across all age classes and grids that comprise each country, predicted incidence rates ranged from 43.7 (95% confidence interval: 0.6 to 591.2) in Zimbabwe to 2,957.8 (95% CI: 20.8 to 4,245.2) in South Sudan per 100,000 person-years. Sub-national heterogeneity was evident with the coefficient of variation at the 20 km × 20 km grid-level ranging from 0.7 to 3.3 and was generally lower in high-incidence countries and widely vary-ing in low-incidence countries.

## Conclusion

Our study provides estimates of 20 km × 20 km incidence rate of typhoid fever across sub-Saharan Africa based on data collected from 2000 through 2020. Increased understanding of the subnational geospatial variation of typhoid fever in Africa may inform more effective intervention programs by better targeting resources to heterogeneously disturbed disease risk.

## Author summary

Typhoid fever remains a significant health challenge in low- and middle-income nations, especially in sub-Saharan Africa. A comprehensive understanding of the disease's geospatial distribution is pivotal for its control—a gap previous studies overlooked. Addressing this, we undertook a study to chart the incidence of typhoid fever throughout sub-Saharan Africa. Using data from 2000 to 2022, we developed high-resolution maps with a granularity of 20 km by 20 km, detailing the spatial distribution of typhoid incidence. Our findings reveal pronounced disparities in typhoid incidence across different geospatial tiers: from local communities to entire nations and regions. Particularly, Western and Eastern Africa registered the highest incidences, with children aged 5–14 years being the most vulnerable. Distinctively, countries such as South Sudan reported alarmingly high figures, whereas Zimbabwe had notably fewer cases. Such insights are indispensable for health policymakers at local, national, and global levels. Pinpointing the areas hardest hit by typhoid allows for a more strategic allocation of resources and interventions. Armed with this data, we're better positioned to fight typhoid effectively and safeguard lives in sub-Saharan Africa.

## Introduction

Typhoid fever is a systemic infection caused by *Salmonella enterica* serovar Typhi (S. Typhi) that is transmitted via water or food contaminated by human feces [1]. Typhoid fever often causes mild symptoms such as fever or weakness but untreated may progress to potentially fatal complications such as intestinal perforation [2,3]. Typhoid fever continues to be a major

public health problem in low- and middle-income countries (LMICs) [4–8]. While being the cornerstone for control of typhoid fever, improving water, sanitation, and hygiene (WASH) and food safety is generally a resource-intensive and long-term goal. Other control strategies such as vaccination, that may protect unvaccinated people as well as vaccine recipients [9], and antimicrobials, that may reduce morbidity and mortality among those with typhoid fever while also reducing fecal carriage and onward transmission [1], may facilitate more immediate impact and near-term control. Limited resources for health mean that global and national policy makers need to identify efficient ways to implement these control strategies, which demands detailed understanding of the distribution of disease.

Typhoid fever is estimated to have caused 14 million cases in 2010 and 11 million cases in 2017 globally [6]. Earlier studies reported somewhat varying estimates: 21 million cases worldwide in 2000 [7]; 27 million (interquartile range: 18–36 million) cases in 2010 [10]; 12 million (95% confidence interval: 10–15 million) or slightly higher in 2010 [4, 5]; and 17.8 million cases in 2015 but with much wider uncertainty intervals (95% credible interval: 6.9–48.4 million) [8].

Despite existing estimates, there are still knowledge gaps regarding the variation of typhoid fever incidence at the country or subnational levels. Understanding the subnational geospatial variation of the disease could lead to more efficient and effective intervention strategies by spatially targeting resources to areas at higher risk. Even for estimates that may agree at the global level, significant variation exists at the national and United Nations (UN) subregional levels [6]. Moreover, studies that estimate sub-national variation often rely on national-level covariates for modeling and prediction, potentially missing important local nuances [8]. In addition, existing studies do not account for observations during the recent typhoid surveillance in sub-Saharan Africa (SSA) and upating the estimates is necessary. Except for one prior study [8], the existing research has primarily examined the disease at the national level, neglecting subnational variations. All estimates save two [5,6] rely on surveillance conducted in <20 sites before 2010, with only three sites from SSA including one from the 1980s.

We sought to address the limitations of existing studies and develop a more granular understanding of geospatial variation of typhoid fever incidence rates. In this study, we focused on SSA for which two multi-country surveillance studies, Typhoid Fever Surveillance in Africa Program (TSAP) conducted in 2012–2014, [11] and Severe Typhoid in Africa (SETA) conducted in 2016–2019 [12,13] provide recent data suggesting substatial burden of typhoid fever in the region. The issue of geospatial variation of typhoid fever is particualrly relevant to SSA because countries like the Democratic Republic of the Congo, Ghana, Malawi, and Zimbabwe are currently rolling out their typhoid vaccination programs [14–16]. We modeled age-stratified incidence rates by integrating data on the incidence rates and catchement area with recently synthesized geospatial covariates that represent factors that potentially influence the transmission of typhoid fever [17–20]. Using the predicted incidence rates on 20 km × 20 km grids of SSA, we characterized the geospatial variation of typhoid fever and potential implications.

#### Methods

We modeled the incidence rate of typhoid fever on  $20 \text{ km} \times 20 \text{ km}$  grids by exploring the association of observed incidence rates with geospatial covariates representing factors that may influence the transmission of *S*. Typhi, such as access to improved water and sanitation, health conditions of the population, or environmental conditions. We chose the resolution of  $20 \text{ km} \times 20 \text{ km}$  because this resolution allows us to explore the subnational heterogeneity with reasonable computational cost. The expected number of  $20 \text{ km} \times 20 \text{ km}$  grids per country

varies from about 28 (e.g., Gambia) to about 5,862 (e.g., Democratic Republic of the Congo). A previous study explored the subnational heterogeneity of the burden of cholera in sub-Saharan Africa at a similar resolution [21]. We used multivariate regression models, assuming the typhoid case count per 100,000 person-years followed either a Poisson or negative binomial distribution. The negative binomial distribution was considered to address overdispersion.

#### Incidence rate data

The data on incidence rates were extracted as a part of a broader project investigating the occurrence of typhoid fever, regardless of whether the instances were documented through case reports, outbreak investigations, or longitudinal surveillance. The literatures search procedure has been detailed elsewhere [22] and only studies that reported incidence rates from longitudinal surveillance were included in this study (Fig 1). In summary, we comprehensively searched through PubMed as well as preprint servers such as Social Science Research Network, bioRxiv, and medRxiv to identify relevant studies published between Jan 1, 2000 and Dec 31, 2022, using the search query "typhoid AND Africa.". Among 301 studies that reported occurrence of typhoid fever, five studies provided information on incidence rates of typhoid fever observed during longitudinal surveillance.

Incidence rates for typhoid fever were available at 22 sites in 13 countries within SSA (Fig 2 and **Table A** in S1 Information) [11,23–29]. Estimates were based on hospital-based passive surveillance with the exception of one study that involved active surveillance of households [28]. All studies used multipliers in their calculation of incidence rates to account for healthcare seeking behavior and the recruitment proportion; some studies used an additional multiplier to account for blood culture sensitivity [24,25,27]. We standardized the observed incidence rates across different studies by using the estimates that accounted for healthcare seeking behavior and recruitment proportion during the surveillance period. We later applied a uniform multiplier of 1/0.6 to modeled incidence rates to account for blood culture sensitivity (estimated to be  $\sim 60\%$ ) [4]. The incidence rates were broken down by age: 0–1, 2–4 yo, 5-14 yo, and >14 yo. Three studies [24,26,27] were not included in the analysis as they did not report typhoid incidence for the relevant age classes. One of these studies presented an aggregated incidence across all age classes [24], while the remaining two studies reported incidence among children under the age of 5 years [26] and children under the 15 years of age [27], respectively. One study used an age category of 5–17 years instead of 5–14 years [28]; we included this study in the model without further adjustment assuming incidence rate observed among those aged 5–17 years would be the same as the one observed in those 5–14 years.

To model incidence rates using geospatial covariates, we linked observed incidence rates from each surveillance site to the geospatial covariates that fall on the catchment area of the relevant healthcare facilities. Catchment areas were based on unpublished catchment area information collected during the surveillance (Section A and **Fig A** in <u>S1 Information</u>) [11] or the descriptions available in the studies [23–28].

#### Geospatial covariates

Variables that may influence the water-borne or food-borne transmission of *S*. Typhi and cover the entire continent of SSA with at least 20 km × 20 km resolution were explored for their predictive capacity for the incidence rate of typhoid fever (**Table 1**). We explored access to safe drinking water and access to improved sanitation facilities [18,30–32] considering the water-borne or food-borne transmission of typhoid fever [1]. We included annual mean temperature [33–35], annual precipitation [34], elevation [36], and distance to water [37] because they may have the potential to influence the transmission of water-borne diseases and have



https://doi.org/10.1371/journal.pntd.0011902.g001

been shown to be associated with an increased risk of typhoid fever. We also explored covariates that showed high predictive capacity in the previous modeling study [8], namely percent paved roads, percent of the population living in extreme poverty, prevalence of stunting, and prevalence of HIV [8]. Because the resolution of these covariates used in the previous modeling study [8] were low (i.e., national or first administrative unit level), we used high-resolution estimates that were recently synthesized: prevalence of stunting, wasting, and underweight under the age of five years [17] and the prevalence of HIV among adults aged 15–49 years [19]. We could not identify higher-resolution covariates for the two covariates—percent paved roads and people living in extreme poverty—and instead used a covariate indicating travel



**Fig 2. Observed typhoid incidence rates in sub-Saharan Africa, 2000–2020.** (A) Year and location of the surveillance. (B) Incidence rates per 100,000 person-years that do not account for the sensitivity of blood culture tests. Points and bars indicate mean and 95% confidence or credible intervals. Points with no bars represent zero incidence rates in which interval estimates were not reported. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html.

time to the nearest urban center [20] assuming all of these covariates partly capture infrastructure. We included human population density [38] as a covariate to reflect that the surveillance was conducted in sites with varying population density [39,40]. We created the population size

Covariate	Description	Spatial resolution*	Temporal resolution* and duration	Ref.	
Access to piped water	Percentage people with access to piped water (on or off premises)	5 km by 5 km	Yearly 2000-2017	[18]	
Improved water	Percentage people with access to other improved drinking water facilities (protected wells and springs, bottled water, rainwater collection, bought water)	5 km by 5 km	Yearly 2000-2017	[ <u>18</u> ]	
Access to surface water	Percentage people with access to surface water	5 km by 5 km	Yearly 2000-2017	[18]	
Access to piped sanitation	Percentage people with access to septic or sewer sanitation	5 km by 5 km	Yearly 2000-2017	[ <u>18</u> ]	
Access to improved sanitation	Percentage access to improved sanitation facilities (e.g., improved latrines, ventilated improved latrines, composting toilets)	5 km by 5 km	Yearly 2000-2017	[ <u>18</u> ]	
Access to open defecation	Percentage people who practiced open defecation	5 km by 5 km	Yearly 2000-2017	[18]	
Annual rainfall	Time mean flux of rain, snow and hail measured as the height of the equivalent liquid water in a square meter per day, which was then summed across the year	1 km by 1 km	Daily 2000–2017	[41]	
Annual mean temperature	Temperature of air at a height of 2 metres above the Earth's surface.	1 km by 1 km	Monthly 2000–2017	[41]	
HIV prevalence	Prevalence of HIV among adults (aged 15-49 years)	5 km by 5 km	Yearly 2000-2017	[19]	
Travel time to the nearest city	Travel time to the nearest urban center	1 km by 1 km	2015	[20]	
Elevation	Elevation data from AWS Open Data Terrain Tiles	~30 m	Various data sources	[36]	
Distance to water	The distance for each pixel to the nearest water cell (inland and sea)	~0.3 km by 0.3 km)	2015	[37]	
Stunting prevalence	Proportion of children (0–59 months) with a height-for-age z-score that is more than two standard deviations below the World Health Organization's median growth reference standards for a healthy population	5 km by 5 km	Yearly 2000–2017	[ <u>17</u> ]	
Wasting	Proportion of children (0–59 months) with a weight-for-height that is more than two standard deviations below the World Health Organization's median growth reference standards for a healthy population	5 km by 5 km	Yearly 2000–2017	[17]	
Underweight	Proportion of children (0–59 months) with a weight-for-age z-score, that is more than two standard deviations below the World Health Organization's median growth reference standards for a healthy population	5 km by 5 km	Yearly 2000–2017	[17]	
Population density	The number of people in a grid cell	1 km by 1 km	Yearly 2000-2017	[38]	

#### Table 1. Characteristics of the candidate covariates included in the model.

\*Covariates were transformed to annual estimates on 20 km by 20 km grids before modeling

https://doi.org/10.1371/journal.pntd.0011902.t001

for each 20 km  $\times$  20 km grids by aggregating the values from 1 km  $\times$  1 km grids using data from the WorldPop project [38].

These geospatial covariates were based on the estimates from previous analyses. For instance, the access to water and sanitation facilities are the mean estimates from the geostatistical modeling analyses that were based on Demographic and Health Survey (DHS), Multiple Indicator Cluster Surveys, and other household surveys and censuses [18]. Similarly, prevalence of stunting, wasting, and underweight under the age of five years [17], and the prevalence of HIV among adults aged 15–49 years [19] were the mean estimates of previous geostatistical modeling analyses of the data from DHS and other sources. Other covariates such as temperature and precipitation came from satellite observations [41]. Elevation data came from 'elevatr' package [36], which uses Terrain Tiles Public Dataset [42]. These covariates were at finer resolutions (e.g.,  $5 \text{ km} \times 5 \text{ km}$  or  $1 \text{ km} \times 1 \text{ km}$ ) and created 20 km  $\times$  20 km grids by aggregating smaller grids and setting the value of the grid as the mean of the comprising smaller grids. Geospatial distribution of the covariates appears in **Figs B-Q** in S1 Information.

We matched the year and location of data collection between observed incidence rates and geospatial covariates. For covariates whose values were available for the period of 2000 through

2017, we assumed the values after 2017 were the same as in 2017. Elevation and distance to water covariates were assumed to be time-invariant. To determine the location for the observed incidence rates, we identified grid cells that were covered by the catchment area during the surveillance (using 'extract' function of raster package in R). A grid cell was defined to be covered by an area if its center was inside the area. The average value of the covariates across the grids was taken as the representative value when multiple grids were covered by the area.

#### Predicting incidence rate

We employed a generalized linear modeling framework to explore the association between the outcome data and geospatial covariates. The outcome was quantified as case per 100,000 person-years for specific age groups (0–1 yo, 2–4 yo, 5–14 yo, and >14 yo) on 20 km  $\times$  20 km grids. Mathematical details about the models appear in Section B of S1 Information. For each model assuming a varying outcome distribution (Poisson or negative binomial), we conducted the following variable selection process. We removed covariates whose variance inflation factors (VIFs) were over ten, or that were highly correlated with other variables (Pearson's r > 0.7) while having low correlation with the outcome (Pearson's r < 0.1) to reduce multicollinearity [43]. We performed a forward stepwise variable selection method based on Akaike Information Criterion (AIC) [44] because AIC criteria can be straightforwardly applied to generalized models, including generalized linear models, non-linear models, and non-normal distributed data. A covariate was included in the model if its introduction led to a decrease in the AIC. Conversely, the covariate was not included in the model if its introduction didn't lead to a decrease in AIC. The covariate with the highest positive or negative correlation with the outcome was tested first, followed by the covariate with the next highest correlation with the outcome, and so on. We further selected covariates based on their *p*-values, their impact on root mean-square error (RMSE) in the leave-one-out (LOO) cross-validation [45] (Section C in S1 Information), and the age-distribution of the predicted incidence rates. We retained variables if their *p*-values were lower than 0.05, their addition reduced the RMSE in the LOO cross-validation, and they produced incidence rate that resembled the data in age distribution (i.e., incidence rates higher in 2-14 yo than in 0-1 yo and >14 yo). We chose the Poisson model for 0-1 yo and the negative binomial models for the other age classes. A list of covariates included in the final model and their associated parameter estimates appear in Table B in S1 Information.

To calculate the predicted number of cases at the grid level, we multiplied the predicted incidence rates with the population size for each 20 km × 20 km grid. Since the WorldPop data lacked age-specific population sizes, we estimated the age-specific population sizes by multiplying the population size at the grid level with the overall proportions of each age class at the country level, sourced from information in the UN World Population Prospectus [46]. To determine the predicted number of cases at the country level, we aggregated the grid-level cases across all the grids within the country of interest. Subsequently, we computed the country-level incidence rate by dividing the number of cases by the population size at the country level. To compare the predicted number of cases with estimates of previous studies, we assumed blood culture sensitivity at 60% as in the previous study [4] and used population size from 2017 [6,8]. We set the upper bound of the predicted incidence rates to 10,000 per 100,000 person-years if the predicted incidence rates were higher than 10,000 per 100,000 person-years.

#### Data sharing

All data and R code required to replicate the figures and tables presented in this paper are accessible in the GitHub repository [47]. Shapefiles specific to the African continent are

available for download from GADM at: <u>https://gadm.org/license.html</u>. Information about the sources for all other publicly available data used in this study is documented in the respective sections where these data are referenced.

### Results

#### Model and predictor selection

The negative binomial regression model yielded the best predictive performance for age groups 2–4 yo, 5–14 yo, and >14 yo, while the Poisson model demonstrated superior performance for the 0–1 yo group (refer to Table C in <u>S1 Information</u> for RMSE values). These models with the best performance were then employed to predict the incidence rates of typhoid for the corresponding age group on 20 km × 20 km grids. Across all models, the covariates measuring the distance to water bodies and the proportion of people using surface water were included, and both demonstrated a positive association with the typhoid incidence rate, but to a varying extent (refer to <u>Table 2</u>). However, different covariates were additionally selected and therefore the final set of covariates varied by age group. For the 0–1 yo age group, population density and open defecation exhibited a positive association, whereas elevation, access to piped sanitation, and wasting showed a negative association with the incidence rate. For the 2–4 yo age group, only population density was additionally included in the model, and it had a positive association. For the 5–14 yo age group, only the proportion of population who have access to piped water was additionally included in the model, and it had a positive association.

#### Comparison between observed and predicted incidence rates

Overall, 50 incidence rate estimates were available for 0-1 yo, 2-4 yo, 5-14 yo, and >14 yo age classes for 22 sites in 13 countries in sub-Saharan Africa (**Table A** in <u>S1 Information</u>). All studies reported incidence rates for all age classes except for one study [26] reporting incidence

Variable	0-1 yo			2-4 yo			5-14 yo			>14 yo		
	Est.	SE	<i>p</i> -value	Est.	SE	p-value	Est.	SE	p-value	Est.	SE	<i>p</i> -value
Improved water				0.034	0.023	0.149						
Improved sanitation												
Annual rainfall										0.691	0.238	0.004
Annual mean temperature												
Stunting prevalence										11.935	4.332	0.006
HIV prevalence												
Travel time to the nearest city												
Elevation	-3.848	0.860	< 0.001									
Distance to water	0.082	0.017	< 0.001	0.088	0.042	0.038	0.064	0.022	0.004	0.101	0.028	0.000
Access to piped water							0.029	0.008	0.000	0.043	0.013	0.001
Access to piped sanitation	-0.112	0.024	< 0.001							0.028	0.013	0.029
Access to surface water	0.079	0.015	< 0.001	0.080	0.039	0.040	0.036	0.018	0.049	0.104	0.024	0.000
Access to open defecation	0.068	0.014	< 0.001									
Wasting	-70.458	11.988	< 0.001									
Underweight												
Population density	1.312	0.210	0.000	0.398	0.259	0.125						

Table 2. The model with the final set of covariates for each group. Blank cells indicate the variables that were excluded during the variable selection process.

Est. = estimates; SE = standard error

https://doi.org/10.1371/journal.pntd.0011902.t002





rates for those who were 4 years old or younger. Surveillance sites included both urban and rural sites and provided incidence rates observed over the period from 2006 through 2019. The surveillance duration for each site ranged from 13 months (East Wad Medani, Sudan for July 2012—July 2013) to >5 years (Asante Akim North, Ghana for March 2010—May 2012 and May 2016—May 2019).

A substantial fraction of the observed incidence rates was predicted by the model in the LOO cross-validation (Fig 3). For the 0–1 yo, 2–4 yo, 5–14 yo, and >14 yo age groups, 13.6%, 50.0%, 53.8%, 62.5% of the observations were within the 95% confidence interval of the predictions, respectively.

#### Predicted incidence rate

Predicted incidence rates varied at the grid, subnational, country, and subregional levels. For instance, variations at the subnational level were evident in all age classes (Fig 4A-4D). Overall, 2–4 yo were predicted to have the highest incidence rates (417.5 [95% CI: 10.7 to 7291.5] per 100,000 person-years) followed by 5–14 yo (308.0 [95% CI: 33.7 to 6590.6]), and > 14 yo (46.2



**Fig 4.** Predicted incidence rates per 100,000 person-years for 0–1 yo (A), 2–4 yo (B), 5–14 yo (C), and >14 yo (D), summarized at subnational levels for 2017. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html.

[95% CI: 1.0 to 5661.3]); 0–1 yo were predicted to have the lowest incidence rates (21.0 [95% CI: 0.0 to 2840.7]) (**Fig R** in <u>S1 Information</u>).

Country-level variation of the predicted incidence is also evident in all age groups (**Fig S** and **Table D** in <u>S1 Information</u>). When aggregated across all age classes and grids that comprise each country, predicted incidence rates ranged from 43.7 (95% CI: 0.6 to 591.2) in Zimbabwe to 2,957.8 (95% CI: 20.8 to 4,245.2) in South Sudan per 100,000 person-years (**Fig 5**, **Table D** in <u>S1 Information</u>). At the UN subregional level [48], Middle Africa had the highest incidence rate (1,017.1 [95% CI: 32.0 to 1,275.6] per 100,000 person-years), followed by Eastern Africa, Western Africa, and Southern Africa (**Table E** in <u>S1 Information</u>).

The expected annual number of cases in the continent adjusted for the 2017 population was 1,883,572 (95% CI: 127,474 to 6,535,604). By UN subregion, the number of typhoid fever cases was highest in Eastern Africa (4,782,220 [95% CI: 277,999 to 11,897,898] cases) followed by Western Africa (3,544,420 [95% CI: 1,904 to 9,874,415]), Middle Africa (2,515,459 [95% CI: 98,917 to 5,479,315]), and Southern Africa (145,130 [95% CI: 1,904 to 749,018]) (**Tables F and G** in <u>S1 Information</u>). Countries with high typhoid incidence (e.g., South Sudan) were associated with the lower coefficient of variation (CV) of 20 km × 20 km grid-level incidence rates whereas low-incidence countries were associated with a wide range of CVs from higher (e.g., Burkina Faso) and lower (e.g., Gambia) (**Fig 5**).



**Fig 5.** Country-level typhoid incidence rates per 100,000 person-years (A) and the coefficient of variation (CV) of 20 km × 20 km grid-level incidence rates (B). Mean country incidence rates varied from 43.7 to 2,957.8 and CV varies from 0.7 to 3.3. Shapefiles specific to the African continent are available for download from GADM at: https://gadm. org/license.html.

#### Discussion

We estimated the age-specific incidence rates of typhoid fever for SSA on 20 km  $\times$  20 km grids using typhoid incidence rate data over the period of 2000 through 2022 including recently completed surveillance [13] and high-resolution geospatial covariates matched to the catchment area for the surveillance sites. Predicted incidence rates showed substantial geospatial variation at all levels: subnational, national and Africa subregional levels. To the best of our knowledge, this study represents the first exploration of the high-resolution distribution of typhoid incidence rates across the African continent, using the most current available data. While extracting actionable insights directly from the study results may pose challenges, achieving a more comprehensive understanding of the geospatial distribution of typhoid burden by age could lead to more effective and efficient targeted intervention programs [49,50].

While our model primarily focuses on predicting incidence rates per 100,000 people rather than providing explanations for the association between covariates and incidence rates, the selected covariates in the model appear to offer reasonable explanations. For instance, using surface water (all age groups), population density (0–1 yo and 2–4 yo age groups), and open defecation (0–1 yo) appear to show a positive association with typhoid incidence per capita. Conversely, covariates like access to piped sanitation and elevation indicate a negative association with the incidence rate, whereas a negative association between the prevalence of wasting and the incidence rate is not intuitively clear.

Existing estimates of the typhoid fever burden generally align on a global scale but exhibit considerable variation at the UN subregional level [6]. Our UN subregional-level estimates, overall, tend to surpass those of Mogasale*et al.*[4] and GBD estimates [6] while closely resembling the estimates presented by Antillón *et al.*[8]. Notably, there's a substantial disparity for Eastern Africa, where our estimates are approximately twice as high as those reported by Antillón *et al.*[8]. (**Tables E** and **G** in <u>S1 Information</u>). However, due to the limited number of observations, both our estimates and those of by Antillón *et al.*[8] display significant uncertainty ranges, which considerably overlap and encompass the estimates of Mogasale *et al.*[4] whereas GBD estimates [6] persistently remain lower without overlapping with other estimates.

Intrepreting agreement or disagreement among estimates should be done with care, however, and needs to account for the context of the models and the data used in each study. For instance, estimates by Mogasale *et al.*[4] and Antillón *et al.*[8] did not account for the data from TSAP and SETA [11, 12], which are a major source of data for incidence rate modeling in our study. Also, the methods used in the studies by Mogasale *et al.*[4] and Kim *et al.*[5] do not use any geospatial covariates to model observed incidence rates except for national-level estimates for the access to improved water source.

Incidence rate estimates from this study were characterized by large uncertainties. One of the main reasons for the wide confidence intervals is the small sample size. While we included all the datasets we found in the peer-reviewed literature including one pre-print as of January 2023, the number of observations was fewer than 30 for all age groups. The 95% confidence intervals of estimates from this study overlapped substantially with those of Antillón *et al.*[8] and encapsulate estimates of other studies (**Table I** and **Fig T** in <u>S1 Information</u>); however, the lower bounds of the confidence intervals tended to be lower than those of other studies with many of them close to zero, which might reflect the fact that the data set we used included zeros in 20% of the data.

Our study has limitations. First, while population-based surveillance provides reliable inputs for the model to predict the incidence of typhoid fever, these data were available from only 22 sites in 13 countries [11,13,23–28], which makes it challenging to infer incidence for all of SSA. While research collecting additional data and developing ways to use other information such as serosurveys [51] should continue, use of the most up-to-date incidence rate data and high-resolution geospatial covariates enable us to produce updated incidence rates at the subnational level. Second, like the estimates of previous studies for typhoid and other diseases [4-8,10,52], our estimates are predictions of annual incidence rates that are based on static and retrospective analyses. While these estimates can provide clues to the current incidence of disease, the burden of typhoid fever is seasonally and secularly variable, and effective policy making would require real-time data synthesis. Third, we set an upper limit for the incidence rates as 10,000 per 100,000 person-years, which is around 4 times higher than the highest of the observed incidence rates, while some of the model estimates at 20 km  $\times$  20 km grids go beyond the limit. This may introduce a bias that incidence rate estimates are lower for some areas. For 0–1 yo, 2–4 yo, 5–14 yo, and >14 yo, 0.2%, 5.6%, 6.2%, 5.1% of the total 60,535 valid grids had estimated values over 10,000 per 100,000 person-years. Fourth, while we used granular geospatial covariates to model the incidence rates, we had to rely on the nationally averaged age distribution to produce the age-specific predicted number of cases. Therefore, age-specific predicted number of cases may either overestimate or underestimate the true number of cases. Fifth, we modeled incidence rates of typhoid fever separately for each age group and this led to different sets of covariates for different age groups. Associations with the covariate in the models should not be interpreted as causal and many vary by age to the extent that certain covariates are not significant predictors of incidence in all age groups. Sixth, the predictive power of our model is limited for 0-1 yo, potentially because a large fraction of observations are zeroes. This warrants additional data and updates.

#### Conclusion

Our research offers the most detailed prediction to date of the typhoid fever incidence rate in Sub-Saharan Africa (SSA), utilizing the latest data on incidence rates and geospatial variables at a resolution of 20 km  $\times$  20 km. This study enhances understanding of the geographical variations in typhoid fever and provides valuable data that could inform targeted intervention strategies for typhoid control in the region.

## **Supporting information**

S1 Information. Table A. Longitudinal surveillance studies of typhoid fever incidence in Africa. Incidence rate data used for modeling come from four published articles and one preprint surveillance study that reported incidence rates measures in sub-Saharan Africa since 2000. Table B. Estimated coefficients of covariates in the proposed model, sub-Saharan Africa, 2017. Greyed cells indicate variables that were removed before modeling to reduce multicollinearity. Cells with the blue background indicate variables that were removed because the p-values were larger than or near to 0.05 and excluding them reduced the LOO cross-validation RMSE. Section A. Catchment area for the incidence rates Fig A. Catchment area represented as  $20 \text{ km} \times 20 \text{ km}$  grids. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Fig B. Percentage access to improved sanitation facilities based on Deshpande et al. [18]. The improved sanitation includes sewer or septic tanks and other improved sanitation facilities (improved latrines, ventilated improved latrines, composting toilets). The dataset is available at the IHME: https://cloud.ihme.washington.edu/s/ bkH2X2tFQMejMxy. Fig C. Percentage access to sewer or septic sanitation facilities based on Deshpande et al. [18]. The dataset is available at the IHME: ttps://cloud.ihme.washington.edu/s/ bkH2X2tFQMejMxy. Fig D. Percentage open defecation based on Deshpande et al. [18]. The dataset is available at the IHME: https://cloud.ihme.washington.edu/s/bkH2X2tFQMejMxy. Fig E. Access to improved drinking water based on Deshpande et al. [18]. The improved water indicates access to piped water according to the JMP definition and includes piped (piped on or off premises) and other improved (protected wells and springs, bottled water, rainwater collection, bought water) water. The dataset is available at the IHME: https://cloud.ihme.washington.edu/ s/bkH2X2tFQMejMxy. Fig F. Access to piped drinking water based on Deshpande et al. [18]. The dataset is available at the IHME: https://cloud.ihme.washington.edu/s/ bkH2X2tFQMejMxy. Fig G. Use of surface water based on Deshpande et al. [18]. The dataset is available at the IHME: https://cloud.ihme.washington.edu/s/bkH2X2tFQMejMxy. Fig H. Precipitation. The values indicate annual precipitation summed across daily precipitation data Climate Hazards group Infrared Precipitation with Stations (CHIRPS) data set [41]. The dataset is available at https://cds.climate.copernicus.eu/cdsapp#!/dataset/insitu-gridded-observationsglobal-and-regional?tab=form. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Fig I. Annual mean temperature for 2017. Monthly mean temperatures [41] were averaged. The dataset is available at https://cds. climate.copernicus.eu/cdsapp#!/dataset/insitu-gridded-observations-global-and-regional?tab= form. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Fig J. Prevalence of stunting among children under the age of 5 in Africa for 2017 based on the study by Kinyoki et al. [17]. The dataset is available at https:// cloud.ihme.washington.edu/index.php/s/Q5CGeazb4iNsDQA. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Fig K. Prevalence of wasting among children under the age of 5 in Africa for 2017 based on the study by Kinyoki et al. [17]. The dataset is available at https://cloud.ihme.washington.edu/index.php/ s/Q5CGeazb4iNsDQA. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Fig L. Prevalence of underweight among children under the age of 5 in Africa for 2017 based on the study by Kinyoki et al. [17]. The dataset is available at https://cloud.ihme.washington.edu/index.php/s/Q5CGeazb4iNsDQA. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/ license.html. Fig M. Prevalence of HIV infection among adults (15-49 years old) in sub-Saharan Africa for 2017 based on the study by Dwyer-Lindgren et al.[19]. The dataset is available at https://ghdx.healthdata.org/record/ihme-data/africa-hiv-prevalence-geospatial-estimates-20002017. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Fig N. Travel time to cities (in minutes) in Africa for 2017 based on the study by Weiss et al. [20]. The dataset is available at https://data.malariaatlas.org/maps. Shapefiles specific to the African continent are available for download from GADM at: https:// gadm.org/license.html. Fig O. Distance to water [37]. The value indicates the distance for each pixel to the nearest water cell (inland and sea) at 20 km × 20 km resolution. For the water cell the distance to water is 0 km. The dataset is available at https://data.ceda.ac.uk/neodc/ globolakes/data/v1/limnology. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Fig P. Elevation [36]. Elevation data come from estimates by Mapzen that combine several digital elevation model (DEM) such as the Shuttle Radar Topography Mission (SRTM), the USGS National Elevation Dataset (NED), Global DEM (GDEM), and others. R package 'elevatr' serves as an API that enables an access to the elevation estimates by Mapzen hosted at Amazon Web Services Terrain Tiles. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/ license.html. Fig Q. Population count per pixel based on the WorldPop [38]. Mosaiced 1km resolution global dataset were aggregated to create 20 km resolution dataset. The dataset is available at https://hub.worldpop.org/geodata/listing?id=64. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Section B. Multivariate regression for Poisson and negative binomial models. Section C. Model validation. Table C. The root mean squared error (RMSE) values of the proposed model (linear regression model) with the competing models (Poisson and Negative Binomial regression models). Fig R. Distribution of predicted incidence rates per 100,000 person years by age (A) and incidence rate ratio (B) with children aged 5-14 years as a reference group. Incidence rates were summarized at subnational levels. Red dots indicate observed incidence rates. Fig S. Predicted incidence rates per 100,000 person-years for 0-1 yo (A), 2-4 yo (B), 5-14 yo (C), and >14 yo (D) summarized at country level. Bold and thin lines inside Africa represent country borders and first-level administrative divisions, respectively. Shapefiles specific to the African continent are available for download from GADM at: https://gadm.org/license.html. Table D. Estimated incidence rates per 100,000 person years by country, sub-Saharan Africa, 2017. Table E. Estimated incidence rate of typhoid fever per 100,000 persons per year by Africa subregion, 2017. Table F. Estimated number of cases by country and age group, sub-Saharan Africa, 2017. Table G. Estimated number of cases by Africa subregion, 2017 (unit = thousands). Table H. Comparison of incidence rates per 100, 000 person-years by country. Fig T. Incidence rate estimates per 100,000 person-years by country. (A) shows existing incidence rate estimates at the country level. The upper bounds of some estimates by Antillón et al. and in the current study go over 4,000. (B) highlights the differences between the estimates by Antillón et al. and by this study that take a similar approach of using grid-level geospatial covariates. (DOCX)

## Acknowledgments

We thank Justin Im (International Vaccine Institute) and John D. Clemens (International Vaccine Institute) for their review and feedback on this article. The authors alone are responsible for the views expressed in this article and they do not necessarily represent the decisions, policy, or views of their affiliated organizations.

## **Author Contributions**

Conceptualization: Jong-Hoon Kim.

Data curation: Jong-Hoon Kim, Chaelin Kim, Gi Deok Pak, Prerana Parajulee.

Formal analysis: Jungsoon Choi.

Funding acquisition: Jong-Hoon Kim, Florian Marks.

- **Investigation:** Jong-Hoon Kim, Vittal Mogasale, Hyon Jin Jeon, Ellis Owusu-Dabo, Raphaël Rakotozandrindrainy, Abdramane Soura Bassiahi, Mekonnen Teferi, Octavie Lunguya-Metila, Florian Marks.
- Methodology: Jong-Hoon Kim, Jungsoon Choi.
- Project administration: Jong-Hoon Kim.
- **Resources:** Jong-Hoon Kim, Andrea Haselbeck, Se Eun Park, Hyon Jin Jeon, Raphaël Rakotozandrindrainy, Abdramane Soura Bassiahi, Mekonnen Teferi, Octavie Lunguya-Metila, Florian Marks.
- Software: Jong-Hoon Kim, Jungsoon Choi.
- Supervision: Jong-Hoon Kim, Florian Marks.
- **Validation:** Jong-Hoon Kim, Jungsoon Choi, Annie J. Browne, Christiane Dolecek, Virginia E. Pitzer, John A. Crump, Simon I. Hay, Florian Marks.
- Visualization: Jong-Hoon Kim, Jungsoon Choi.
- Writing original draft: Jong-Hoon Kim, Jungsoon Choi.
- Writing review & editing: Chaelin Kim, Gi Deok Pak, Prerana Parajulee, Andrea Haselbeck, Se Eun Park, Vittal Mogasale, Hyon Jin Jeon, Annie J. Browne, Ellis Owusu-Dabo, Raphaël Rakotozandrindrainy, Abdramane Soura Bassiahi, Mekonnen Teferi, Octavie Lunguya-Metila, Christiane Dolecek, Virginia E. Pitzer, John A. Crump, Simon I. Hay, Florian Marks.

#### References

- Crump JA, Sjölund-Karlsson M, Gordon MA, Parry CM. Epidemiology, Clinical Presentation, Laboratory Diagnosis, Antimicrobial Resistance, and Antimicrobial Management of Invasive Salmonella Infections. Clin Microbiol Rev. 2015; 28: 901–937. https://doi.org/10.1128/CMR.00002-15 PMID: 26180063
- Birkhold M, Coulibaly Y, Coulibaly O, Dembélé P, Kim DS, Sow S, et al. Morbidity and Mortality of Typhoid Intestinal Perforation Among Children in Sub-Saharan Africa 1995–2019: A Scoping Review. World J Surg. 2020; 44: 2892–2902. https://doi.org/10.1007/s00268-020-05567-2 PMID: 32430740
- Butler T, Knight J, Nath SK, Speelman P, Roy SK, Azad MA. Typhoid fever complicated by intestinal perforation: a persisting fatal disease requiring surgical management. Rev Infect Dis. 1985; 7: 244–256. https://doi.org/10.1093/clinids/7.2.244 PMID: 3890097
- Mogasale V, Maskery B, Ochiai RL, Lee JS, Mogasale VV, Ramani E, et al. Burden of typhoid fever in low-income and middle-income countries: a systematic, literature-based update with risk-factor adjustment. The Lancet Global Health. 2014; 2: e570–e580. https://doi.org/10.1016/S2214-109X(14)70301-8 PMID: 25304633
- Kim J-H, Mogasale V, Im J, Ramani E, Marks F. Updated estimates of typhoid fever burden in sub-Saharan Africa. Lancet Glob Health. 2017; 5: e969. https://doi.org/10.1016/S2214-109X(17)30328-5 PMID: 28911760
- Stanaway JD, Reiner RC, Blacker BF, Goldberg EM, Khalil IA, Troeger CE, et al. The global burden of typhoid and paratyphoid fevers: a systematic analysis for the Global Burden of Disease Study 2017. Lancet Infect Dis. 2019; 19: 369–381. https://doi.org/10.1016/S1473-3099(18)30685-6 PMID: 30792131
- Crump JA, Luby SP, Mintz ED. The global burden of typhoid fever. Bull World Health Organ. 2004; 82: 346–353. PMID: 15298225

- Antillón M, Warren JL, Crawford FW, Weinberger DM, Kürüm E, Pak GD, et al. The burden of typhoid fever in low- and middle-income countries: A meta-regression approach. PLoS Negl Trop Dis. 2017; 11: e0005376. https://doi.org/10.1371/journal.pntd.0005376 PMID: 28241011
- Qadri F, Khanam F, Liu X, Theiss-Nyland K, Biswas PK, Bhuiyan AI, et al. Protection by vaccination of children against typhoid fever with a Vi-tetanus toxoid conjugate vaccine in urban Bangladesh: a cluster-randomised trial. Lancet. 2021; 398: 675–684. <u>https://doi.org/10.1016/S0140-6736(21)01124-7</u> PMID: 34384540
- Buckle GC, Walker CLF, Black RE. Typhoid fever and paratyphoid fever: Systematic review to estimate global morbidity and mortality for 2010. J Glob Health. 2012; 2: 010401. <u>https://doi.org/10.7189/jogh.02.</u> 010401 PMID: 23198130
- Marks F, von Kalckreuth V, Aaby P, Adu-Sarkodie Y, El Tayeb MA, Ali M, et al. Incidence of invasive salmonella disease in sub-Saharan Africa: a multicentre population-based surveillance study. Lancet Glob Health. 2017; 5: e310–e323. https://doi.org/10.1016/S2214-109X(17)30022-0 PMID: 28193398
- Park SE, Toy T, Cruz Espinoza LM, Panzner U, Mogeni OD, Im J, et al. The Severe Typhoid Fever in Africa Program: Study Design and Methodology to Assess Disease Severity, Host Immunity, and Carriage Associated With Invasive Salmonellosis. Clin Infect Dis. 2019; 69: S422–S434. <u>https://doi.org/10. 1093/cid/ciz715 PMID: 31665779</u>
- Marks F, Im J, Park SE, Pak G, Jeon H, Wandji Nana LR, et al. The Severe Typhoid in Africa Program: Incidences Of Typhoid Fever in Burkina Faso, Democratic Republic of Congo, Ethiopia, Ghana, Madagascar, And Nigeria. Rochester, NY; 2022. https://doi.org/10.2139/ssrn.4292849
- Duff N. Liberia Becomes First Country in Africa to Introduce Typhoid Conjugate Vaccine. In: Take on Typhoid [Internet]. [cited 8 Aug 2022]. Available: https://www.coalitionagainsttyphoid.org/typhoid-news/ liberia-becomes-first-country-in-africa-to-introduce-typhoid-conjugate-vaccine/
- Lunguya PO, Biomédicale IN de R, team the T study. First TCV effectiveness study in Central Africa launched in the DRC. In: Take on Typhoid [Internet]. 14 Feb 2022 [cited 8 Aug 2022]. Available: <u>https://</u> www.coalitionagainsttyphoid.org/drc-launches-tcv-study/
- Africa's first-ever mass typhoid fever vaccination campaign ends in Zimbabwe. In: Take on Typhoid [Internet]. [cited 8 Aug 2022]. Available: https://www.coalitionagainsttyphoid.org/typhoid-news/africasfirst-ever-mass-typhoid-fever-vaccination-campaign-ends-in-zimbabwe/
- Kinyoki DK, Osgood-Zimmerman AE, Pickering BV, Schaeffer LE, Marczak LB, Lazzar-Atwood A, et al. Mapping child growth failure across low- and middle-income countries. Nature. 2020; 577: 231–234. https://doi.org/10.1038/s41586-019-1878-8 PMID: 31915393
- Deshpande A, Miller-Petrie MK, Lindstedt PA, Baumann MM, Johnson KB, Blacker BF, et al. Mapping geographical inequalities in access to drinking water and sanitation facilities in low-income and middleincome countries, 2000–17. The Lancet Global Health. 2020; 8: e1162–e1185. <u>https://doi.org/10.1016/</u> S2214-109X(20)30278-3 PMID: 32827479
- Dwyer-Lindgren L, Cork MA, Sligar A, Steuben KM, Wilson KF, Provost NR, et al. Mapping HIV prevalence in sub-Saharan Africa between 2000 and 2017. Nature. 2019; 570: 189–193. <u>https://doi.org/10.1038/s41586-019-1200-9 PMID: 31092927</u>
- Weiss DJ, Nelson A, Gibson HS, Temperley W, Peedell S, Lieber A, et al. A global map of travel time to cities to assess inequalities in accessibility in 2015. Nature. 2018; 553: 333–336. <u>https://doi.org/10.1038/nature25181 PMID: 29320477</u>
- Lessler J, Moore SM, Luquero FJ, McKay HS, Grais R, Henkens M, et al. Mapping the burden of cholera in sub-Saharan Africa and implications for control: an analysis of data across geographical scales. Lancet. 2018; 391: 1908–1915. https://doi.org/10.1016/S0140-6736(17)33050-7 PMID: 29502905
- Kim J-H, Parajulee P, Nguyen TT, Wasunkar S, Mogasale V, Park SE, et al. Occurrence of human infection with Salmonella Typhi in sub-Saharan Africa. medRxiv; 2023. p. 2023.09.22.23295884. https://doi.org/10.1101/2023.09.22.23295884
- Thriemer K, Ley B, Ame S, von Seidlein L, Pak GD, Chang NY, et al. The burden of invasive bacterial infections in Pemba, Zanzibar. PLoS One. 2012; 7: e30350. <u>https://doi.org/10.1371/journal.pone.</u> 0030350 PMID: 22363426
- Crump JA, Youssef FG, Luby SP, Wasfy MO, Rangel JM, Taalat M, et al. Estimating the Incidence of Typhoid Fever and Other Febrile Illnesses in Developing Countries. Emerg Infect Dis. 2003; 9: 539– 544. https://doi.org/10.3201/eid0905.020428 PMID: 12737736
- Srikantiah P, Girgis FY, Luby SP, Jennings G, Wasfy MO, Crump JA, et al. Population-based surveillance of typhoid fever in Egypt. Am J Trop Med Hyg. 2006; 74: 114–119. PMID: 16407354
- 26. Nielsen MV, Sarpong N, Krumkamp R, Dekker D, Loag W, Amemasor S, et al. Incidence and characteristics of bacteremia among children in rural Ghana. PLoS One. 2012; 7: e44063. <u>https://doi.org/10. 1371/journal.pone.0044063</u> PMID: 22970162

- Guiraud I, Post A, Diallo SN, Lompo P, Maltha J, Thriemer K, et al. Population-based incidence, seasonality and serotype distribution of invasive salmonellosis among children in Nanoro, rural Burkina Faso. PLOS ONE. 2017; 12: e0178577. https://doi.org/10.1371/journal.pone.0178577 PMID: 28692655
- Breiman RF, Cosmas L, Njuguna H, Audi A, Olack B, Ochieng JB, et al. Population-based incidence of typhoid fever in an urban informal settlement and a rural area in Kenya: implications for typhoid vaccine use in Africa. PLoS One. 2012; 7: e29119. <u>https://doi.org/10.1371/journal.pone.0029119</u> PMID: 22276105
- 29. Meiring JE, Shakya M, Khanam F, Voysey M, Phillips MT, Tonks S, et al. Burden of enteric fever at three urban sites in Africa and Asia: a multicentre population-based study. Lancet Glob Health. 2021; 9: e1688–e1696. https://doi.org/10.1016/S2214-109X(21)00370-3 PMID: 34798028
- Brockett S, Wolfe MK, Hamot A, Appiah GD, Mintz ED, Lantagne D. Associations among Water, Sanitation, and Hygiene, and Food Exposures and Typhoid Fever in Case–Control Studies: A Systematic Review and Meta-Analysis. The American Journal of Tropical Medicine and Hygiene. 2020; 103: 1020– 1031. https://doi.org/10.4269/ajtmh.19-0479 PMID: 32700668
- Tadesse BT, Khanam F, Ahmed F, Im J, Islam MT, Kim DR, et al. Prevention of typhoid by Vi conjugate vaccine and achievable improvements in household WASH: Evidence from a cluster-randomized trial in Dhaka, Bangladesh. Clin Infect Dis. 2022; ciac289. <u>https://doi.org/10.1093/cid/ciac289</u> PMID: 35412603
- 32. Im J, Khanam F, Ahmmed F, Kim DR, Kang S, Tadesse BT, et al. Prevention of Typhoid Fever by Existing Improvements in Household Water, Sanitation, and Hygiene, and the Use of the Vi Polysaccharide Typhoid Vaccine in Poor Urban Slums: Results from a Cluster-Randomized Trial. Am J Trop Med Hyg. 2022; 106: 1149–1155. https://doi.org/10.4269/ajtmh.21-1034 PMID: 35385827
- Liang M, Ding X, Wu Y, Sun Y. Temperature and risk of infectious diarrhea: a systematic review and meta-analysis. Environ Sci Pollut Res Int. 2021. <u>https://doi.org/10.1007/s11356-021-15395-z</u> PMID: 34268683
- **34.** Gao Q, Liu Z, Xiang J, Zhang Y, Tong MX, Wang S, et al. Impact of Temperature and Rainfall on Typhoid/Paratyphoid Fever in Taizhou, China: Effect Estimation and Vulnerable Group Identification. Am J Trop Med Hyg. 2021; 106: 532–542. https://doi.org/10.4269/ajtmh.20-1457 PMID: 34872055
- Chua PLC, Ng CFS, Tobias A, Seposo XT, Hashizume M. Associations between ambient temperature and enteric infections by pathogen: a systematic review and meta-analysis. Lancet Planet Health. 2022; 6: e202–e218. https://doi.org/10.1016/S2542-5196(22)00003-1 PMID: 35278387
- Hollister J, Shah T, Robitaille AL, Beck MW, Johnson M. elevatr: Access Elevation Data from Various APIs. 2021. https://doi.org/10.5281/zenodo.5809645
- Carrea L, Embury O, Merchant CJ. Datasets related to in-land water for limnology and remote sensing applications: distance-to-land, distance-to-water, water-body identifier and lake-centre co-ordinates. Geoscience Data Journal. 2015; 2: 83–97. https://doi.org/10.1002/gdj3.32 PMID: 28616229
- Lloyd CT, Sorichetta A, Tatem AJ. High resolution global gridded data for use in population studies. Sci Data. 2017; 4: 170001. https://doi.org/10.1038/sdata.2017.1 PMID: 28140386
- von Kalckreuth V, Konings F, Aaby P, Adu-Sarkodie Y, Ali M, Aseffa A, et al. The Typhoid Fever Surveillance in Africa Program (TSAP): Clinical, Diagnostic, and Epidemiological Methodologies. Clin Infect Dis. 2016; 62 Suppl 1: S9–S16. https://doi.org/10.1093/cid/civ693 PMID: 26933028
- 40. Carey ME, MacWright WR, Im J, Meiring JE, Gibani MM, Park SE, et al. The Surveillance for Enteric Fever in Asia Project (SEAP), Severe Typhoid Fever Surveillance in Africa (SETA), Surveillance of Enteric Fever in India (SEFI), and Strategic Typhoid Alliance Across Africa and Asia (STRATAA) Population-based Enteric Fever Studies: A Review of Methodological Similarities and Differences. Clin Infect Dis. 2020; 71: S102–S110. https://doi.org/10.1093/cid/ciaa367 PMID: 32725221
- **41.** Copernicus Climate Change Service. Temperature and precipitation gridded data for global and regional domains derived from in-situ and satellite observations. 2021 [cited 20 Jul 2022]. Available: https://cds.climate.copernicus.eu/doi/10.24381/cds.11dedf0c
- Terrain Tiles v1.1 Update · Mapzen. [cited 2 Aug 2023]. Available: https://www.mapzen.com/blog/ terrain-tiles-v1.1/
- Shrestha N. Detecting Multicollinearity in Regression Analysis. Am J Appl Math. 2020; 8: 39–42. <a href="https://doi.org/10.12691/ajams-8-2-1">https://doi.org/10.12691/ajams-8-2-1</a>
- Yamashita T, Yamashita K, Kamimura R. A Stepwise AIC Method for Variable Selection in Linear Regression. Communications in Statistics—Theory and Methods. 2007; 36: 2395–2403. https://doi.org/ 10.1080/03610920701215639
- **45.** Arlot S, Celisse A. A survey of cross-validation procedures for model selection. Statistics Surveys. 2010; 4: 40–79. https://doi.org/10.1214/09-SS054

- United Nations. World Population Prospects 2022. Available: <a href="https://population.un.org/wpp/Download/Standard/Population/">https://population.un.org/wpp/Download/Standard/Population/</a>
- 47. Kim J-H. Mapping the incidence rate of typhoid fever in sub-Saharan Africa. Available: <a href="https://github.com/kimfinale/TyphoidMapping">https://github.com/kimfinale/TyphoidMapping</a>
- United Nations Statistics Division–Standard Country and Area Codes Classifications. Available: <a href="https://unstats.un.org/unsd/methodology/m49/">https://unstats.un.org/unsd/methodology/m49/</a>
- Date KA, Bentsi-Enchill A, Marks F, Fox K. Typhoid fever vaccination strategies. Vaccine. 2015; 33: C55–C61. https://doi.org/10.1016/j.vaccine.2015.04.028 PMID: 25902360
- Getchell M, Mantaring EJ, Yee K, Pronyk P. Cost-effectiveness of sub-national geographically targeted vaccination programs: A systematic review. Vaccine. 2023; 41: 2320–2328. <u>https://doi.org/10.1016/j.vaccine.2023.02.006</u> PMID: 36781333
- Aiemjoy K, Seidman JC, Saha S, Munira SJ, Sajib MSI, Sium SMA, et al. Estimating typhoid incidence from community-based serosurveys: a multicohort study. The Lancet Microbe. 2022; 3: e578–e587. https://doi.org/10.1016/S2666-5247(22)00114-8 PMID: 35750069
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496: 504–507. https://doi.org/10.1038/nature12060 PMID: 23563266